Accelerator Status Report

Run Schedule: March 11, 2010 to July 22, 2010
June 16, 2010
Injector    Energy 56.62 MeV
· Sometimes QE measurement tool does not work, and the readbacks of 0L02BCM and I to V hang up. The Gigatronics power meter needs to be power cycled.

· ROT ½ wave plate readback on the TOOLS screen does not update. We can use the command “caput” until the problem is fixed.

· Wien angle was changed for Hall A Sunday and rolled back Monday.
· (6/8) Wien was flipped for Hall A.

· (6/8) 100 kV window comparator adjusted to clear the GUN HV fault.

Accelerator   Linac Energy 503.25 MeV
· The accelerator was running well with some hardware failures.
· RF separators still trip off sometimes.

· (6/13) The shunt module of MAB6T03 was replaced. ??? (2.5 hrs.)
· (6/12) VBV1L12A micro-switch failed and was replaced (1.5 hrs.)

· (6/10) Thursday the Load Shed Test was successfully performed from 1130 to 1500. It took more than five hours to resume beam delivery after the test due to problems with magnets (Arc1 and Arc7) and viewer system. 

· (After an extended down or some settings of RF or magnets may change) we should always make sure the settings are where they were prior to sending beam. 
· The accelerator was running with scheduled and unscheduled downtimes.
· RF separators tripped a lot. EES has been trying to resolve the issue.
· (6/6) Sunday the electronics package of the injector Aperture # 1 cooling water system was water damaged, which caused about three hours of downtime.
· (6/6) iocse10 unresolved due to a failed battery, which caused Hall A more than two hours of beam time.
· (6/4 and 6/5) Over the weekend the CHL tripped twice causing about three shifts of downtime. The CHL personnel are still investigating the causes of the trips. 
· (6/3) An Accelerator Hurricane Preparedness Power Test was conducted successfully on Thursday: the generator started properly; all Automatic Transfer Switches operated properly; the CHL did not trip. Issues with some systems were identified and will be addressed accordingly. It took more than two shifts to prepare and conduct the test, and restore the beam delivery.

· (6/2) RF zone 1L13 was bypassed. It was arcing in the cabinet. The zone was recovered on June 3.

Hall A---Pass 1, Prex from March 11 to May 24, 2010
· PREx will end Monday (6/21). APEX will kick in. There will be beam line change.
· Hall A target has issues with its chamber vacuum and one of its dipoles. 
· Hall A BCM cavity temperature not stable sometimes.

· Hall A Compton noise level was lowered.

· Hall A started to take beam owl shift of June 4.

Hall B ---Pass 5, g9-Frost from March 11 to July 22, 2010
· The hall reported that sometimes there is a current spike during Hall A current ramping. 
· Hall B beam was fine.

· Hall B needs about a shift to polarize its target every Tuesday.

Hall C---QWeak from June 8 to August 15, 2010
· Hall C start-up has been delayed to the week of the 4th of July.

· You must be trained before you can be a Hall C lead sweeper. 

· QWeak start-up will be delayed again for a few days. Not sure when yet.

